Author's response to reviews

Title: Factors that lead to the use of crack cocaine in combination with marijuana in Brazil: a qualitative study. Janaina R. Goncalves1 and Solange A. Nappo2*
*Corresponding author: Solange A. Nappo solangenappo@gmail.com Janaina R. Goncalves jana.ru.bio@hotmail.com 1,2Brazilian Center of Information on Psychotropic Drugs (CEBRID), Department of Preventive Medicine, Universidade Federal de Sao Paulo, Sao Paulo, Brazil. Rua Botucatu, 740, CEP 040-23062.

Authors:

Janaina R Gonçalves (jana.ru.bio@hotmail.com)
Solange Aparecida SA Nappo (solangenappo@gmail.com)

Version: 2 Date: 21 September 2014

Author's response to reviews:

1. The pages and lines were numbered.
2. The approval of Ethics Committee was mentioned at the end of Methodology session.
3. The authors' identification was moved on the title page
4. The name of second author was corrected